Author's response to reviews

Title: Induction of cell cycle arrest at G1 and S phases and cAMP-dependent differentiation in C6 glioma by low concentration of cycloheximide

Authors:

Xijun Liu (xliu3@psu.edu)
Jin-Ming Yang (juy16@psu.edu)
Samual S Zhang (ssz3@psu.edu)
Xinyuan Liu (xyliu@sibs.ac.cn)
David X Liu (dliu@psu.edu)

Version: 6 Date: 6 December 2010

Author's response to reviews: see over
Dec. 6, 2010

Dear Editor,

Thank you for accepting our ms (Manuscript ID 2039941156378699) for publication in BMC Cancer. We had made final corrections/adjustments according to your instructions, which are detailed below:

1. Title: please ensure that the title is identical in the manuscript and the submission system. Currently they do not match.
   We had deleted the extra word “cell” from the submission system so that it is now identical to the title in the ms.

2. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name'(e.g. Joe F Bloggs). Currently they do not match.
   We replaced the Initial name “Shao-Min” with “S” in the ms so that it conforms with system format.

3. Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.
   Addresses were added in the ms as requested.

4. Authors’ contributions: Please indicate that all authors read and approved the final manuscript.
   Yes, all authors read and approved the final ms.

5. Figures: Figure legend 1 contains graphics. These will either need to be included within the image as a key, or written in text (e.g. "black square"). The keys are in the image and therefore it seems to have satisfied the requirement.

6. Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

   References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.
   List of full names are given and “et al.” removed.

7. Figures 2 and 4: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
   We remove some space from both Figs 2 and 4.
8. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
We carefully re-examined the final documents and made sure they are error-free.

Thank you.

David Liu
Psnn State University
Pennsylvania, USA